

# Constraining Uncertainty in Projected Gross Primary Production With Machine Learning

Manuel Schlund, Veronika Eyring, Gustau Camps-valls, Pierre Friedlingstein,

Pierre Gentine, Markus Reichstein

## ▶ To cite this version:

Manuel Schlund, Veronika Eyring, Gustau Camps-valls, Pierre Friedlingstein, Pierre Gentine, et al.. Constraining Uncertainty in Projected Gross Primary Production With Machine Learning. Journal of Geophysical Research: Biogeosciences, 2020, 125 (11), 10.1029/2019jg005619. hal-03094573

## HAL Id: hal-03094573 https://hal.sorbonne-universite.fr/hal-03094573

Submitted on 4 Jan 2021

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



ſ

# **JGR** Biogeosciences

## **RESEARCH ARTICLE**

10.1029/2019JG005619

#### **Key Points:**

- An emergent constraint on CO<sub>2</sub> seasonal cycle amplitude changes reduces uncertainties in global mean gross primary production projections
- A machine learning model with multiple predictors can further constrain the spatial distribution of gross primary production
- High-latitude ecosystems show higher gross primary production increase over the 21st century compared to regions closer to the equator

#### **Supporting Information:**

Supporting Information S1

#### Correspondence to:

M. Schlund, manuel.schlund@dlr.de

#### **Citation**:

Schlund, M., Eyring, V., Camps-Valls, G., Friedlingstein, P., Gentine, P., & Reichstein, M. (2020). Constraining uncertainty in projected gross primary production with machine learning. *Journal of Geophysical Research: Biogeosciences*, *125*, e2019JG005619. https://doi.org/10.1029/2019JG005619

Received 19 DEC 2019 Accepted 18 SEP 2020 Accepted article online 27 OCT 2020

#### $\textcircled{\sc c}2020.$ The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## Constraining Uncertainty in Projected Gross Primary Production With Machine Learning

Manuel Schlund<sup>1</sup> D, Veronika Eyring<sup>1,2</sup> D, Gustau Camps-Valls<sup>3</sup> D, Pierre Friedlingstein<sup>4,5</sup>, Pierre Gentine<sup>6,7</sup> D, and Markus Reichstein<sup>8,9</sup> D

<sup>1</sup>Deutsches Zentrum für Luft- und Raumfahrt (DLR), Institut für Physik der Atmosphäre, Oberpfaffenhofen, Germany, <sup>2</sup>Institute of Environmental Physics (IUP), University of Bremen, Bremen, Germany, <sup>3</sup>Image Processing Laboratory (IPL), Universitat de València, València, Spain, <sup>4</sup>College of Engineering, Mathematics and Physical Sciences, University of Exeter, Exeter, UK, <sup>5</sup>LMD/IPSL, ENS, PSL Université, Ecole Polytechnique, Institut Polytechnique de Paris, Sorbonne Université, CNRS, Paris, France, <sup>6</sup>Department of Earth and Environmental Engineering, Columbia University, New York, NY, USA, <sup>7</sup>Earth Institute and Data Science Institute, Columbia University, New York, NY, USA, <sup>8</sup>Department of Biogeochemical Integration, Max Planck Institute for Biogeochemistry, Jena, Germany, <sup>9</sup>Michael-Stifel-Center Jena for Data-driven and Simulation Science, Jena, Germany

**Abstract** The terrestrial biosphere is currently slowing down global warming by absorbing about 30% of human emissions of carbon dioxide (CO<sub>2</sub>). The largest flux of the terrestrial carbon uptake is gross primary production (GPP) defined as the production of carbohydrates by photosynthesis. Elevated atmospheric CO<sub>2</sub> concentration is expected to increase GPP ("CO<sub>2</sub> fertilization effect"). However, Earth system models (ESMs) exhibit a large range in simulated GPP projections. In this study, we combine an existing emergent constraint on CO<sub>2</sub> fertilization with a machine learning approach to constrain the spatial variations of multimodel GPP projections. In a first step, we use observed changes in the CO<sub>2</sub> seasonal cycle at Cape Kumukahi to constrain the global mean GPP at the end of the 21st century (2091-2100) in Representative Concentration Pathway 8.5 simulations with ESMs participating in the Coupled Model Intercomparison Project Phase 5 (CMIP5) to  $171 \pm 12$  Gt C yr<sup>-1</sup>, compared to the unconstrained model range of 156–247 Gt C yr $^{-1}$ . In a second step, we use a machine learning model to constrain gridded future absolute GPP and gridded fractional GPP change in two independent approaches. For this, observational data are fed into the machine learning algorithm that has been trained on CMIP5 data to learn relationships between present-day physically relevant diagnostics and the target variable. In a leave-one-model-out cross-validation approach, the machine learning model shows superior performance to the CMIP5 ensemble mean. Our approach predicts an increased GPP change in northern high latitudes compared to regions closer to the equator.

**Plain Language Summary** About a quarter of human emissions of carbon dioxide  $(CO_2)$  is absorbed by vegetation and soil on the Earth's surface and hence does not contribute to global warming caused by  $CO_2$  in the atmosphere. Thus, in order to better define the remaining carbon budgets left to meet particular warming targets like the 1.5°C of the Paris Agreement, it is important to accurately quantify the carbon uptake by plants in the future. Currently, this is modeled by Earth system models yet with great uncertainties. In this work, we present an alternative machine learning approach to reduce spatial uncertainties of vegetation carbon uptake in future climate projections using observations of today's conditions.

## 1. Introduction

The response of the terrestrial carbon cycle to changes in atmospheric  $CO_2$  and climate is a major source of uncertainty in climate projections (Bodman et al., 2013; Booth et al., 2012; M. Collins et al., 2013). Future terrestrial carbon sensitivity is mainly driven by two feedback mechanisms: the concentration-carbon and the climate-carbon feedbacks (M. Collins et al., 2013; Friedlingstein et al., 2006; Gregory et al., 2009). The first one is connected to the  $CO_2$  fertilization effect (Walker et al., 2020) where elevated atmospheric  $CO_2$  concentrations increase photosynthesis rates and generally lead to a higher terrestrial carbon uptake, thus constituting a negative feedback. The second one, the climate-carbon feedback, is driven by temperature and precipitation changes leading to a smaller land



carbon uptake because of increased temperature and water stress on photosynthesis and higher ecosystem respiration costs, as well as increased fire frequency.

The terrestrial biosphere currently absorbs about 30% of the total anthropogenic  $CO_2$  emissions (Friedlingstein et al., 2019), yet whether that benefit will perdure in the future remains unclear (Friedlingstein et al., 2006). Accurate prediction of future carbon uptake is, however, of paramount importance to accurately determine the allowable  $CO_2$  emissions to meet particular temperature targets, such as the 1.5°C of the Paris Agreement (Matthews et al., 2009; UNFCCC, 2015). One way to quantify future terrestrial carbon uptake is via the use of land surface models within Earth system models (ESMs). Here we use ESM outputs from the fifth phase of the Coupled Model Intercomparison Project (CMIP5) (Taylor et al., 2012). However, since this model ensemble shows a large spread in the evolution of the land carbon sink (Ciais et al., 2013), a careful statistical evaluation and refinement of the multimodel mean and its uncertainty is required.

A prominent method to constrain future model estimates using observations is the so-called "emergent constraint" approach (Allen & Ingram, 2002), which is based on an intermodel relationship between an observable quantity in the past climate and a projected quantity of the future climate. Emergent constraints have already been successfully applied to carbon cycle processes (Cox et al., 2013; Wenzel et al., 2014; Wenzel, Cox, et al., 2016) and are thought to be a promising technique to reduce uncertainties in climate model ensemble output (Eyring et al., 2019). However, two limitations of this technique are the use of a single (globally or regionally averaged) variable per climate model and the assumption of a linear relationship between the observed and the target variable. Other methods involve a weighting of the multimodel average based on model performance relative to observations (Knutti et al., 2017; Sanderson et al., 2017), including process-oriented approaches like the multiple diagnostic ensemble regression (MDER) (Karpechko et al., 2013; Senftleben et al., 2019; Wenzel, Eyring, et al., 2016).

In this paper, we introduce a new two-step approach that utilizes aspects of both aforementioned techniques in combination with a supervised machine learning algorithm to constrain uncertainties in multimodel projections of gross primary production (GPP) with observations. In the first step, we apply an existing emergent constraint on CO<sub>2</sub> fertilization (Wenzel, Cox, et al., 2016) to constrain ESMs' responses to rising atmospheric  $CO_2$  concentration using observations of the increase of the  $CO_2$  seasonal cycle amplitude at Cape Kumukahi, Hawaii (Keeling et al., 2005). In a second step, we introduce a supervised machine learning algorithm based on boosting trees (Friedman, 2001) to learn an empirical spatial relationship that links grid-wise future GPP to historical processes relevant to its simulation under present-day conditions. In combination with observational products of the predictors, that relationship can be used to further constrain uncertainties in the projected spatial maps of GPP at the end of the 21st century in the Representative Concentration Pathway (RCP) 8.5 scenario (Riahi et al., 2011). We examine both constraining the absolute GPP and the fractional change in GPP as two independent approaches and target variables. Unlike univariate linear regression used in the MDER algorithm, the proposed Gradient Boosted Regression Tree (GBRT) algorithm is able to handle multiple predictors and copes with nonlinearities in the data. GBRT is a well-known and successful tool used for interpolation, classification and prediction in other fields of data science and engineering (De'ath, 2007; Elith et al., 2008). In the context of climate science, GBRT was recently applied to identify the key drivers of spatial variations of the ratio of plant transpiration to total terrestrial evapotranspiration in ESMs (Lian et al., 2018).

In this study, we combine an emergent constraint approach with the GBRT algorithm to reduce uncertainties in projected GPP. Section 2 provides an overview of the methods and data used in this paper. The results are presented in section 3, and section 4 closes with a summary and discussion.

#### 2. Methods and Data

Here we briefly describe the methods and data used in this study. Section 2.1 reviews the method of a previously published emergent constraint on  $CO_2$  fertilization from Wenzel, Cox, et al. (2016) that we apply here to our study to rescale the climate model output (Step 1). This includes a discussion of the caveats that are associated with this emergent constraint that attributes changes in the observed increase in the  $CO_2$  seasonal cycle amplitude at Cape Kumukahi entirely to the increase in atmospheric  $CO_2$  concentration, yet





**Figure 1.** Schematic illustration of our two-step approach. In Step 1 an emergent constraint by Wenzel, Cox, et al. (2016) is used to constrain the global mean fractional GPP change over the 21st century. Moreover, this constraint is used to rescale two different gridded target variables: absolute GPP at the end of the 21st century and fractional GPP change over the 21st century. In Step 2, a machine learning model is used to constrain these two target variables (Step 2a: absolute GPP; Step 2b: fractional GPP change) in two independent approaches.

other studies provide indications that part of this increase is due to climate and land use changes (Bastos et al., 2019; Forkel et al., 2016; Piao et al., 2018; Zhao et al., 2016). In section 2.2 the machine learning technique used in Step 2 to constrain gridded GPP projections is presented. Figure 1 shows an overview of our two-step approach. Section 2.3 describes how uncertainty and predictive skill are quantified.

A complete list of all CMIP5 models used in this study is given in supporting information Table S1. To apply the emergent constraint in Step 1, we need emission-driven CMIP5 simulations. The number of ESMs included is therefore seven, as in Wenzel, Cox, et al. (2016). More details on the methods and the experimental setup are given as supporting information.

#### 2.1. Constraining the CO<sub>2</sub> Fertilization Effect (Step 1)

In this section, we present the observational emergent constraint on the CO<sub>2</sub> fertilization effect by Wenzel, Cox, et al. (2016) which we use to globally rescale the models for their individual bias in the sensitivity of GPP to an increase in atmospheric CO<sub>2</sub> concentration. Our objective here is to present a general methodology and suggest further improvements upon this constraint that could be used in the future. Wenzel, Cox, et al. (2016) showed a correlation between the increase in the  $CO_2$  seasonal cycle amplitude simulated by models and the projected  $CO_2$ -fertilization. When combined with the observed trends in the  $CO_2$  seasonal cycle amplitude, the fractional GPP change at the time of CO<sub>2</sub> doubling relative to preindustrial conditions in a biogeochemically coupled simulation where the atmospheric  $CO_2$  concentration increases by 1% per year (1%BGC) was constrained. The predictor in this study is the sensitivity of the CO<sub>2</sub> seasonal cycle amplitude to rising atmospheric  $CO_2$  concentrations, defined as the slope of the linear regression between the  $CO_2$ seasonal cycle amplitude and the annual mean atmospheric CO<sub>2</sub> concentrations (see Figure S1 for details). The emergent constraint is physically motivated by the hypothesis that increasing terrestrial GPP is a main driver for the observed changes in the CO<sub>2</sub> seasonal cycle amplitude (Gray et al., 2014; Keeling et al., 1996; Zhao & Zeng, 2014). This hypothesis is based on the fact that the seasonal cycle in atmospheric CO<sub>2</sub> concentration originates from photosynthesis and decomposition processes: in summer, photosynthesis removes  $CO_2$  from the atmosphere; in winter, additional  $CO_2$  is added to the atmosphere due to the decomposition

of organic matter (Keeling et al., 1995). Thus,  $CO_2$  fertilization leads to an increase of the  $CO_2$  seasonal cycle amplitude due to increased  $CO_2$  removal from the atmosphere. We note that this emergent constraint implicitly assumes that there is no temperature-driven increase in respiration over time. Therefore, this approach most likely overestimates the true sensitivity of GPP to an increase in the atmospheric  $CO_2$  concentration.

However, we note that this emergent constraint is not undisputed. In other studies, the observed change in the CO<sub>2</sub> seasonal cycle amplitude has been attributed to other factors. Using a dynamic global vegetation model, Forkel et al. (2016) find that the increase in the CO<sub>2</sub> seasonal cycle amplitude above 40°N is mainly driven by the response of plants to global warming instead of CO<sub>2</sub>. Similarly, Piao et al. (2018) show that elevated atmospheric CO<sub>2</sub> concentration and climate change equally contribute to the increase in the CO<sub>2</sub> seasonal cycle in northern high latitudes with additional smaller contributions from air-sea carbon fluxes and land use changes. For their analysis, Piao et al. (2018) used CO<sub>2</sub> records from 26 stations and nine terrestrial ecosystem models. An evaluation of two atmospheric inversions and 11 land surface models shows contrasting effects of CO<sub>2</sub> fertilization (positive) and global warming (negative) on the CO<sub>2</sub> seasonal cycle amplitude increase in northern high latitudes with negligible contributions from land use changes (Bastos et al., 2019). A further study by Zhao et al. (2016) finds that for seven out of nine dynamic vegetation models, the CO<sub>2</sub> fertilization effect is the strongest driver for the observed changes in the CO<sub>2</sub> seasonal cycle amplitude, while the two remaining models equally attribute this to CO<sub>2</sub> fertilization, climate change and land use changes. Moreover, Winkler, Myneni, and Brovkin (2019) discuss limitations of carbon cycle emergent constraints related to the calculation of the predictor from observation-based data and the intermodel relationship. As a case study, they adopt a further emergent constraint by Winkler, Myneni, Alexandrov, and Brovkin (2019) that uses the response of the leaf area fraction to ambient  $CO_2$  instead of the  $CO_2$  seasonal cycle amplitude sensitivity to constrain the GPP increase in the northern high latitudes. Furthermore, it has been shown that the CMIP5 models (Graven et al., 2013) and the TRENDY dynamic vegetation models (Thomas et al., 2016) greatly underestimate long-term changes in the CO<sub>2</sub> seasonal cycle which questions their ability to attribute changes in the  $CO_2$  seasonal cycle to different contributors. Nevertheless, we emphasize that we here present a generic methodology and use the results of an existing emergent constraint. Our proposed framework is universal and could be based on other emergent constraints in the future.

In contrast to Wenzel, Cox, et al. (2016), instead of idealized simulations we apply the presented emergent constraint on the emission-driven RCP 8.5 simulation (Riahi et al., 2011) and examine whether it still holds for this simulation where forcings other than  $CO_2$  also change. Moreover, we consider the global mean GPP change instead of the GPP change in the northern extratropics as in Wenzel, Cox, et al. (2016). Due to the strong annual mixing of  $CO_2$  in the atmosphere, the change in the  $CO_2$  seasonal cycle amplitude at Cape Kumukahi is representative for the whole globe. Thus, we apply this emergent constraint to the global mean GPP change. However, we note that this approach might introduce errors since the  $CO_2$  fertilization effect could be different for tropical and extratropical ecosystems due to nutrient limitations and the temperature dependence of Rubisco kinetics (Crafts-Brandner & Salvucci, 2000) in the tropics. Nevertheless, we argue that we can use the constraint if the emergent relationship still holds within our considered framework (the RCP 8.5 scenario simulated by the CMIP5 models).

Therefore, our target variable in Step 1 is the global mean GPP change over the 21st century f in the RCP 8.5 scenario, which is calculated from the 10 year mean GPP at the end of the 21st century in the emission-driven fully coupled RCP 8.5 simulations (2091–2100) and the 10 year mean GPP at the end of the 20th century (1991–2000) in the emission-driven historical simulation:

$$f = \frac{GPP(2091-2100)}{GPP(1991-2000)} - 1 \tag{1}$$

This variable is given as a percentage, where negative values correspond to a decrease in GPP, positive values to an increase in GPP and a value of 0% to no GPP change. The  $CO_2$  amplitude sensitivity is calculated for the years 1979–2019, which is the full available time range for the observed  $CO_2$  at Cape Kumukahi (Keeling et al., 2005). Similar to Wenzel, Cox, et al. (2016), we extract the observed  $CO_2$  amplitude from smoothed atmospheric  $CO_2$  concentrations (using a stiff cubic spline function plus a four-harmonics functions with linear gain) as provided by Keeling et al. (2005). For the CMIP5 models,



the respective grid cell closest to Cape Kumukahi (19.5°N, 154.8°W) is extracted. Additionally, the emission-driven historical simulations are extended with the emission-driven RCP 8.5 simulations for the years 2006 to 2019. More details including an illustration of this calculation are shown in Figure S1.

Since the global constraint above cannot be used directly as a predictor in the machine learning approach which operates on gridded values (see section 2.2), we use it as first step to globally rescale the ESM output. This important first step allows us to constrain the large uncertainty in the global  $CO_2$  fertilization effect of the CMIP5 models. In this study, two different gridded target variables are examined: the monthly climatology of future absolute GPP at the end of the 21st century (2091–2100) in the RCP 8.5 scenario (see Step 2a) and the fractional GPP change over the 21st century as defined in Equation 1 (see Step 2b). The global rescaling of the gridded target variables works as follows: let  $f_m$  be the global mean fractional GPP change over the 21st century of ESM m, f' the constrained value of the global mean fractional GPP change and  $y_{m,i}$  the target variable of ESM m. Then, we define the rescaled target variable  $y'_{m,i}$  by

$$y'_{m,i} = y_{m,i} \cdot \frac{f'}{f_m},\tag{2}$$

where the index *i* counts over all grid cells (and months when absolute GPP is used as target variable in Step 2a) and *m* refers to the ESM ( $m \in \{1, 2, ..., 7\}$ ). Thus, values for the individual grid cells/months are rescaled by a factor that is determined by the ratio between the models' global mean GPP change and the constrained global mean GPP change derived from the Wenzel, Cox, et al. (2016) emergent constraint using the observed change in the CO<sub>2</sub> cycle amplitude. Models that simulate a too high global mean GPP change are nudged toward smaller values and vice versa. The rescaled target variables are then fed into the machine learning model (Step 2) presented in the following section. We note that this global rescaling assumes that the relative biases in the fractional GPP change of the individual ESMs are constant over the whole globe.

#### 2.2. Gradient Boosted Regression Trees (Step 2)

For Step 2 we use the GBRT algorithm to find an optimal regression function F relating the target variable y to a set of K predictors  $\mathbf{x} = (x^{(1)}, x^{(2)}, ..., x^{(K)})$  ("features" or "covariates") such that the predicted target variable is given by  $\hat{y} = F(x)$  (De'ath, 2007; Elith et al., 2008; Friedman, 2001, 2002). Its basic elements are decision trees, which use binary splits of the input data to create decision rules. Due to their simple nature, decision trees are easy to interpret but cannot be used to create satisfying predictions for very complex data sets. This issue can be overcome by a technique called "boosting" (Freund & Schapire, 1996). Boosting iteratively improves the performance of "weak learners" (in our case decision trees) in the following way: At first, a single decision tree is fitted to a randomly selected subsample of the training data  $\{(x_i, y_i)\}$  (using a predefined loss function, e.g., the mean square error). Then, further decision trees are linearly added to the regression function F (scaled by a regularization parameter  $v \ll 1$  called "learning rate") and the model is fitted again in the same manner. This ensures that the prediction for poorly modeled training points in the beginning will gradually improve during the training. The whole procedure is repeated until the loss function does not improve anymore, as evaluated on an independent validation data set. Relevant hyperparameters of this algorithm (maximum number of decision trees, maximum depth of the decision trees, learning rate, and the subsample ratio) are optimized by fivefold cross-validation using the training data. More details on the GBRT algorithm can be found in Text S2.

A convenient way to evaluate and interpret GBRT models (or other statistical models in general) is to identify the most relevant predictors. Globally, this so-called "feature importance" is closely related to the number of appearances of a feature in the different decision trees that set up the GBRT model (Friedman, 2001). To evaluate the feature importance regionally (for the different grid cells), we use a model distillation technique called LIME ("local interpretable model-agnostic explanations"), which approximates the GBRT model with a local linear regression model for each grid cell (Ribeiro et al., 2016). The corresponding linear

coefficients  $b_k$  for each feature  $x^{(k)}$  constitute a good proxy to the model's Jacobian  $\frac{\partial y}{\partial x^{(k)}}$ , which is difficult to

compute explicitly for this particular machine learning model. Thus, features with high (low) absolute  $b_k$  have a high (low) impact on the local prediction of the target variable *y*.



As already described in the previous section, we analyze two different target variables in this study: the monthly climatology of absolute GPP at the end of the 21st century (Step 2a) and the fractional GPP change over the 21st century (Step 2b). A schematic representation of Step 2a is shown in Figure 2. The target variable *y* for the machine learning model is the rescaled monthly climatology of future GPP at the end of the 21st century (2091–2100) in the RCP 8.5 scenario given by Equation 2. We relate this target variable to several process-oriented diagnostics *x* of the past climate (historical simulation) listed in Table 1. A single training point (*x<sub>i</sub>*, *y<sub>i</sub>*) for the supervised machine learning algorithm corresponds to the set of the aforementioned diagnostics evaluated on a single grid cell/month of a single climate model (see colored dots in Figures 2a and 2b). Step 2b differs from Step 2a only in the target variable used. Both steps are independent approaches and are evaluated separately.

The main part of Steps 2a and 2b each consist of two phases. In the first phase (training of the algorithm), we build a statistical model based on the empirical relationship between observable process-oriented variables of the present-day climate and future GPP in the CMIP5 ensemble (see gray plane in Figures 2a and 2b). This has a flavor similar to the MDER algorithm, which is a climate model weighting method that uses a multivariate linear regression algorithm to constrain multimodel climate projections of a target variable with observations (Karpechko et al., 2013; Senftleben et al., 2019; Wenzel, Eyring, et al., 2016). However, in contrast to the original MDER algorithm, our multivariate GBRT approach is able to handle multiple predictors with nonlinear dependencies. Moreover, instead of considering a single globally or regionally averaged value per climate model, our approach uses spatiotemporally gridded data for every ESM in the ensemble. This dramatically increases the available number of training data (even if they are not statistically independent) and uniquely enables the statistical model to identify and exploit regional processes. Furthermore, the large training database allows us to perform extensive out-of-sample testing, which is crucial to ensure the model robustness.

In the second phase (predicting with the trained GBRT model), we use observational data products to constrain the RCP 8.5 CMIP5 projection of GPP. To do this, observational data for all predictors  $\mathbf{x}$  of the historical climate (for every grid cell/month) are fed into the trained GBRT model in order to get a spatially constrained projection of GPP (see black dots in Figure 2b). Similar to the MDER algorithm, our approach assumes that there exists an intermodel relationship between the predictors and the target variable which also holds for the true climate. This assumption may seem weak at first glance, but compared to the assumption of traditional weighting approaches (models that are better in simulating the present-day climate are also better in simulating the future climate and vice versa), the MDER assumption is more reasonable and allows correcting models instead of just weighing (Karpechko et al., 2013). Following other MDER studies (Karpechko et al., 2013; Senftleben et al., 2019; Wenzel, Eyring, et al., 2016), we also select diagnostics as predictor variables  $\mathbf{x}$  that are related to GPP-relevant processes (see Table 1 for detailed physical explanations of the links between the predictors and GPP).

The raw CMIP5 and observation-driven data sets are preprocessed before entering the machine learning algorithm (see Text S1). In particular, increases greater than 300% in the fractional GPP change that is used as target variable in Step 2b are removed. These unrealistically high values (making up about 5.9% of the data) occur in places with very small absolute GPP values in the historical simulation that are used in the denominator of the derivation of the fractional GPP change (see Equation 1). In addition, we randomly select 25% of the training data and use it as hold-out test data. This data set is excluded from the training and can therefore be used for an independent validation of our statistical model.

#### 2.3. Assessment of Uncertainty and Predictive Skill

The estimation of the standard prediction error (SPE) for the GBRT models needs to consider three sources of uncertainty: the uncertainty of the machine learning model itself, the uncertainty in the rescaling of the target variable (see Equation 2) and the uncertainty in the input data, that is, errors in the observation-based products. First, the SPE of the GBRT model itself (i.e., uncertainty in the position of the gray plane in Figures 2a and 2b) is estimated as the root-mean-square error of predictions (RMSEP), that is, the root-mean-square error (RMSE) between the predicted and true values of the independent hold-out test data set (Bishop, 2006). This approach is justified when the distribution of residuals is unbiased (see Text S4), in which case the RMSEP equals the standard deviation of the residuals (which we interpret as error). Figure S2 shows that this is approximately correct.



10.1029/2019JG005619



**Figure 2.** Schematic illustration of our machine learning approach to constrain projected absolute GPP (Step 2a). (a) In the first phase of the algorithm (training phase), the model is fitted to the training data interpolating the empirical (nonlinear) relationship between two process-oriented diagnostics of the past climate  $\{x^{(1)}, x^{(2)}\}$  and (rescaled) future GPP (gray plane). The dots show the training points for the supervised machine learning algorithm, each of them representing a single grid cell/month of a single climate model (the different colors correspond to different climate models). (b) In the second phase, observation-based values of the diagnostic (black dots) are fed into the fitted machine learning model to constrain GPP for every grid cell/month to a value which best agrees with the observations. (c) For an independent validation of our method, we use an out-of-sample testing setup based on a leave-one-model-out cross-validation approach, that is, by testing on one climate model left out. For this, we create a data set with known ground truth by excluding a single climate model from the ensemble in the training phase and using it as input in the prediction phase. This allows us to evaluate the root-mean-square error between the predicted and the true values, yielding the so-called root-mean-square error of prediction (RMSEP). By repeating this whole process for all climate models, we create a distribution of RMSEPs and compare it to different statistical models. The schematic illustration of Step 2b differs only in the target variable used (fractional GPP change instead of absolute GPP).



Table 1

Process-Oriented Diagnostics ("Predictors" or "Features") Used in the GBRT Model to Predict the Target Variables

| Name | Description                               | Observation-driven data                          | Used time range | Physical connection to GPP   |
|------|---|--|-----------------|--|
| GPP  | Gross primary productivity                | FLUXNET-MTE (Jung et al., 2011), version May12   | 1991-2000       | -  |
| LAI  | Leaf area index                           | LAI3g (Zhu et al., 2013), version 1 (March 2017) | 1982–2005       | The leaf area index is a measure for the<br>number of leaves in a grid cell. The<br>photosynthesis rate is highly<br>dependent on the number of leaves<br>(and vegetation in general). |
| PR   | Precipitation                             | CRU (Harris et al., 2014), version 4.02          | 1901-2005       | Water is essential for the chemical processes of photosynthesis.   |
| RSDS | Downwelling solar radiation<br>at surface | ERA-Interim (Dee et al., 2011), version 2.0      | 1979–2005       | Solar radiation is essential for the chemical processes of photosynthesis.   |
| Т    | Near-surface air temperature              | CRU (Harris et al., 2014), version 4.02          | 1901-2005       | Near-surface air temperature and<br>photosynthesis rate have a common<br>driver (incoming solar radiation).  |

*Note.* For Step 2a (target variable: absolute GPP), all listed variables are monthly climatologies of the specified time ranges in the historical climate. For Step 2b (target variable: fractional GPP change), the temporal mean over the specified time ranges is calculated for all variables.

The SPE of the GBRT model can be understood as combined error due to the internal variability of the climate system, the different model responses within the climate model ensemble to forcing and the incomplete description of the climate system with the limited number of variables used as input features. Second, the uncertainty in the rescaling is simply assessed as the uncertainty given by the emergent constraint from Step 1. Finally, the uncertainty in the observation-based data (i.e., uncertainty in the position of the black dots in Figure 2b) is assessed by error propagation. The necessary sensitivities of the target variable to the various predictors for each grid cell/month are given by the linear coefficients  $b_k$  given by the LIME approach introduced in section 2.2. All sources of uncertainty are assumed independent and are combined by adding the squared errors, yielding a single estimate of the SPE for each grid cell/month. Further details on the calculation of errors are given in Text S3.

The predictive power of the GBRT algorithm and its robustness is assessed using a leave-one-model-out cross-validation approach (see Figure 2c). This is also known as "pseudoreality" or "model-as-truth" approach in climate sciences (de Elia et al., 2002; Karpechko et al., 2013). For this, a single climate model is removed from the multimodel ensemble and considered as being the true climate. Our statistical model is then fitted on all remaining climate models and a prediction for the "true" model is created. This allows a simple evaluation of the predictive power by calculating the RMSEP. The whole process is repeated for every climate model of the ensemble in order to get a distribution of RMSEPs. Since this approach is not limited to GBRT, it allows a simple comparison of the predictive skill of different statistical models. We compare the GBRT model to the simple CMIP5 multimodel mean and a least absolute shrinkage and selection operator (LASSO) model. The LASSO model (Tibshirani, 1996) is an extended linear regression model that regularizes the least squares solution with a L1-penalty term. In contrast to the ordinary least square regression, the LASSO model promotes sparse solutions (i.e., only a few nonzero coefficients/weights), which acts as an intrinsic feature selection and acts as an effective way to combat overfitting induced by collinearity.

#### 3. Results

#### 3.1. Constraining Global Mean GPP Projections (Step 1)

In Wenzel, Cox, et al. (2016), the observed atmospheric CO<sub>2</sub> concentration at Cape Kumukahi, Hawaii (KUM; 19.5°N, 154.8°W) (Keeling et al., 2005) was used to constrain the GPP change in the 1%BGC run resulting from a doubling of atmospheric CO<sub>2</sub> concentrations to  $32 \pm 9\%$  for extratropical ecosystems (30–90°N). As mentioned in section 2.1, we argue that due to the strong annual mixing of CO<sub>2</sub> in the atmosphere, this emergent constraint can also be applied to the global mean GPP change if the emergent relationship holds. In order to test this, we first use the idealized CMIP5 simulations that were analyzed by Wenzel, Cox, et al. (2016) to calculate the emergent constraint (Figure 3a). The relationship is statistically significant at a 5% significance level ( $R^2 = 0.79, p = 0.007$ ) and predicts a constrained global mean GPP change of





**Figure 3.** Emergent relationship between the global mean fractional change in GPP and the sensitivity of the CO<sub>2</sub> seasonal cycle amplitude to rising atmospheric CO<sub>2</sub> concentrations observed at Cape Kumukahi, Hawaii (KUM). Colored markers refer to CMIP5 models, the orange line and shaded area to the linear regression fit and its corresponding standard prediction error, and the black dashed lines to the observational constraint. (a) Similar to Wenzel, Cox, et al. (2016): The global mean fractional GPP change after CO<sub>2</sub> doubling in 1%BGC simulations defined as 10 year mean of the 1%BGC run centered at the time of CO<sub>2</sub> doubling relative to the 10 year mean of preindustrial control conditions at the beginning of the 1%BGC run. The sensitivity of the CO<sub>2</sub> amplitude for the CMIP5 models is calculated from the years 1860–2005, the corresponding observational value from the years 1979–2019. The constrained global mean GPP change after CO<sub>2</sub> doubling in the 1%BGC run is  $30 \pm 9\%$ . (b) Fractional global mean GPP change over the 21st century calculated from the 10 year mean GPP at the end of the 21st century (2091–2100) in the emission-driven fully coupled RCP 8.5 simulations and the 10 year mean GPP at the end of the 20th century (1991–2000) in the emission-driven fully coupled and observations for better comparability (the historical CMIP5 simulations are extended with the RCP 8.5 simulations for the years 1979–2019. The constrained global mean GPP change over the 21st century is  $39 \pm 7\%$ .

 $30 \pm 9\%$ , which is consistent with the findings of Wenzel, Cox, et al. (2016). This result gives us confidence that the emergent relationship holds for the global mean GPP. Details on the mathematical derivation of the constrained range are given for example in Cox et al. (2013).

In addition to applying it to the idealized simulations, we apply the emergent constraint to the RCP 8.5 scenario. In the 1%BGC simulation, the CO<sub>2</sub> fertilization effect is the only driver of future GPP change since the carbon cycle components of the ESMs only see the increasing CO<sub>2</sub> concentration. However, since the magnitude of the carbon-concentration feedback is believed to be four times the size of the carbon-climate feedback (Gregory et al., 2009), the CO<sub>2</sub> fertilization effect is still the dominant driver of GPP change in the fully coupled simulations (Huntzinger et al., 2017). This is also supported by Wenzel, Cox, et al. (2016), who showed that the GPP increase due to rising atmospheric CO<sub>2</sub> concentrations in the fully coupled historical simulations is proportional to the same quantity in the 1%BGC run. Additional lines of evidence suggest that the modeled CO<sub>2</sub> sensitivity of photosynthesis is the main source of uncertainty of future GPP (Arora et al., 2013; Haverd et al., 2020; Rogers et al., 2017). Thus, the sensitivity of the CO<sub>2</sub> amplitude can be used to approximately constrain global mean GPP change over the 21st century in RCP 8.5, which is shown in Figure 3b. Assuming a significance level of 5%, the emergent relationship is statistically significant ( $R^2 = 0.65$ , p = 0.029).

Using the emergent relationship, the global mean GPP change over the 21st century can be constrained to  $39 \pm 7\%$  (standard error), which is at the lower end of the original CMIP5 range (31–57%). Moreover, the best estimate 39% is also slightly lower than the CMIP5 multimodel mean (43%). The observed GPP averaged over the years 1991–2000 from the FLUXNET-MTE product is  $123 \pm 6$  Gt C yr<sup>-1</sup> (Jung et al., 2011). Assuming independent errors and using Gaussian error propagation, the constrained fractional change corresponds to a global mean GPP of  $171 \pm 12$  Gt C yr<sup>-1</sup> (standard error) at the end of the 21st century (2091–2100) in the RCP 8.5 scenario. In contrast to that, the unconstrained CMIP5 ensemble has a model range of 156-247 Gt C yr<sup>-1</sup> for future GPP in the RCP 8.5 scenario (2091–2100). The resulting global mean GPP change is used to rescale the gridded output of the different climate models with Equation 2. The following two sections show the results for the rescaled absolute future GPP (Step 2a) and the rescaled fractional GPP change (Step 2b) further constrained by the GBRT model.

#### 3.2. Constraining Gridded Absolute Gridded Absolute GPP Projections (Step 2a)

In the second step, the objective is to constrain the spatial distribution of the projected GPP. We use the grid-wise monthly mean climatology of absolute GPP at the end of the 21st century (2091–2100) as target





**Figure 4.** (a) Boxplot of the root-mean-square error of prediction (RMSEP) distributions for six different statistical models used to predict future absolute GPP (Step 2a) using a leave-one-model-out cross-validation approach. The distribution for each statistical model contains seven points (black dots, one for each climate model used as truth) and is represented in the following way: The lower and upper limit of the blue boxes correspond to the 25% and 75% quantiles, respectively. The central line in the box shows the median, the black "x" the mean of the distribution. The whiskers outside the box represent the range of the distribution. Compared to the CMIP5 multimodel mean (MMM) and its corresponding rescaled version (rMMM), the prediction uncertainty measured by the mean RMSEP is significantly reduced by up to 48% and 39%, respectively, when using other statistical models. Moreover, the nonlinear GBRT models can slightly reduce the mean RMSEP compared to the linear LASSO models by about 2% for GBRT-1D (using historical GPP as single predictor) and 3% for the full GBRT model (using all predictors). (b) Relative global feature importance for the different features used in the GBRT model to predict future absolute GPP (Step 2a). The red bars correspond to positive Pearson correlation coefficients between all predictors and the target variable. Due to its strong positive linear relationship with the future GPP, the historical GPP is by far the most important predictor in the model. A local feature importance map (using LIME) is not shown here because GPP is the dominant predictor for all grid cells.

variable, rescaled using Equation 2 to reduce the large uncertainties in the  $CO_2$  fertilization effect in the models. In the first part, we evaluate the GBRT model using the leave-one-model-out cross-validation approach and compare it to other statistical approaches and the unweighted multimodel mean. In the second part, we use observation-based products to predict the target variable.

# 3.2.1. Prediction Error in a Leave-One-Model-Out Cross-Validation Approach and Feature Importance

To get a detailed insight into the performance of our GBRT model, we compare it to five other statistical models: the unweighted CMIP5 multimodel mean of future GPP (MMM), its rescaled version using Equation 2 (rMMM), a linear LASSO regression model using all features as defined in Table 1 (LASSO), a single-predictor LASSO model (LASSO-1D) and a single-predictor GBRT model (GBRT-1D). Both "single-predictor" models use only the historical GPP as feature. The predictive power (in terms of RMSEP) of each statistical model is assessed using the leave-one-model-out cross-validation approach (see section 2.3). This allows us to create an RMSEP distribution for each statistical approach, where the different points of the distributions refer to different training/prediction data set combinations generated by the leave-one-model-out cross-validation approach. Thus, each distribution consists of seven points, one for each climate model. Figure 4a shows the RMSEP distributions for the six different statistical models.

In terms of raw predictive power, the simple MMM is outperformed by every other model. Its prediction uncertainty expressed as mean RMSEP can be reduced by more than 48% by using other statistical models. However, this is not surprising: in contrast to the other statistical models, the simple RCP 8.5 MMM does not take further evidence in form of observations of the historical climate into account. Step 1 of our algorithm can reduce this mean RMSEP by 15% (rMMM) due to the single rescaling of the gridded climate model output with the global mean GPP constraint on the CO<sub>2</sub> fertilization effect. However, since there is a considerably large GPP range in the individual climate models themselves, this reduction is rather small for the gridded values. A far larger reduction of the RMSEP can be achieved by using the regression models LASSO-1D, LASSO, GBRT-1D and GBRT. All of them share similar RMSEP distributions, which can be explained as follows: The historical GPP is strongly (near linearly) correlated to the rescaled end-of-century GPP (with pattern correlation of  $R^2 = 0.83$  for the whole multimodel ensemble). Because of that, all



regression models are able to considerably reduce the RMSEPs compared to the MMM and rMMM. This also shows that the GBRT models successfully learned the linear connection between past and future GPP. Moreover, the nonlinear GBRT models can slightly reduce the mean RMSEP even further compared to the linear LASSO models by about 2% for GBRT-1D and 3% for the full GBRT model. This leads to the conclusion that in addition to the strong linear relationship between past and future GPP, there are small nonlinear relations between the predictors and the target variable that can be used to further reduce the RMSEP. Since the full GBRT model with access to all predictors shows the minimal mean RMSEP, we argue that it is beneficial for our approach to use the multivariate nonlinear GBRT model instead of a linear model because only the GBRT algorithm is able to make use of all features and exploit more complex relationships.

The influence of the different predictors can be further analyzed by evaluating the global feature importance by using the whole training data set and the already trained model. This is shown in Figure 4b. The past GPP with its strong positive linear correlation to future GPP is the most relevant predictor of the trained model with a relative importance of approximately 95%. All other features show values of less than 2%. The global feature importance determines the expected impact of a feature in predictions of the target variable. In our GBRT model the prediction input data of historical GPP mainly determines the machine learning model's prediction of the future GPP in the RCP 8.5 scenario at the end of the 21st century. However, we emphasize that this is only valid for our specific setup (i.e., which variables are used, which algorithm is used, which climate models are considered, etc.). For another problem setup, the global feature importance could change significantly. Since the historical GPP is the most important predictor for all grid cells, we do not show the plot of the local feature importance using LIME. We can get further insights into our GBRT model by analyzing the residuals of the prediction using an independent test data set. The resulting plots show that the algorithm is not overfitting the training data and that the prediction errors on the whole are approximately unbiased (see Figure S2a).

#### 3.2.2. Observation-Based GBRT Prediction of Absolute GPP

In this section, we use the different statistical models to predict absolute GPP at the end of the 21st century. For this, we feed observation-based data (see Table 1) into the regression models GBRT, GBRT-1D, LASSO, and LASSO-1D.

The result of Step 1 on the gridded data is illustrated in Figure 5a, which shows the ratio between rMMM and MMM. This ratio is almost constant over the whole globe, which is not surprising due to the global nature of the rescaling (see Equation 2). The use of the global emergent constraint predicts a slightly lower GPP increase over the 21st century (39%) than the unweighted CMIP5 ensemble mean (43%). The corresponding ratio 39 % /43 %  $\approx$  0.91 is approximately equal to the mean value of rMMM/MMM (0.92). Consequently, the global estimate of rMMM gives a lower GPP at the end of the century (179 Gt C yr<sup>-1</sup>) as MMM (198 Gt C yr<sup>-1</sup>), which can be interpreted as a correction of the ESMs' overestimation of future GPP to changes in the atmospheric CO<sub>2</sub> concentration.

The spatial result of our GBRT model in Step 2a can be visualized by comparing rMMM and the output of the GBRT model. Figure 5b shows the bias of the rescaled CMIP5 RCP 8.5 multimodel mean compared to our GBRT predicted GPP at the end of the 21st century (2091–2100), while Figure 5c shows the bias of the historical CMIP5 multimodel mean compared to the observational FLUXNET-MTE product (Jung et al., 2011) averaged for the period 1991-2000. As reported by Anav et al. (2013), the historical CMIP5 ensemble mean overestimates GPP in most regions, leading to a global mean of 138 Gt C yr<sup>-1</sup>, which is larger than the 123 Gt C yr<sup>-1</sup> estimate for the FLUXNET-MTE product. Regions where GPP is largely overestimated are the western parts of South America, central and southern Africa, and East Asia. On the contrary, GPP is underestimated in small areas of Central America and northern parts of South America. The bias patterns in Figures 5b and 5c are very similar (pattern correlation of  $R^2 = 0.88$ ), which means that the GBRT algorithm detects regional biases in the historical simulation and corrects them in its future predictions. This is also illustrated in Figure 6b, in which many values are close to 1, while only a few negative values exist. However, we emphasize that our approach is only to first order a bias correction (i.e., subtracting the historical bias from the RCP 8.5 multimodel mean). There are clear differences between Figures 5b and 5c, which are also illustrated in Figures 6a and 6b. A simple bias correction would only show constant values for the whole globe, whereas our approach applies different corrections (in sign and magnitude) for different regions. On the global domain, our approach predicts a GPP at the end of the century of 169 Gt C yr<sup>-1</sup>,





**Figure 5.** (a) Ratio of the rescaled CMIP5 ensemble mean of the absolute GPP at the end of the 21st century (2091–2100) in the RCP 8.5 scenario using Equation 2 (rMMM) and its unweighted version (MMM). The plot shows an almost constant value over the whole globe with a mean of 0.92, which corresponds to the ratio of the constrained global mean GPP change over the 21st century (39%) and the CMIP5 ensemble mean global mean GPP change (43%) from Step 1. All values close to 0 for the data set in the denominator have been masked to avoid divisions by 0. (b) Bias between rMMM and our GBRT prediction of for the end of the 21st century. This corresponds to Step 2a of our approach. (c) Bias between the modeled GPP in the CMIP5 multimodel mean of the historical simulation and the FLUXNET MTE observation-based estimate of GPP (red color). Panels (b) and (c) show similar bias patterns (pattern correlation of  $R^2 = 0.88$ ). Thus, the GBRT prediction in Step 2a is able to correct the historical bias of the CMIP5 ensemble relative to the FLUXNET-MTE product.

which is consistent with the global constraint from Step 1 ( $171 \pm 12$  Gt C yr<sup>-1</sup>). In summary, our approach first corrects the CMIP5 models response to CO<sub>2</sub> (Step 1) and second corrects the historical bias of the ESMs relative to observations (Step 2a).

Further illustrations of these results including the uncertainties can be found in Figures S3–S5. Analogous to the results from the leave-one-model-out cross-validation approach, the uncertainties on grid cell level are significantly reduced for the GBRT prediction compared to the multimodel mean results. These errors consider all three sources of uncertainty that we presented in section 2.3 (error in the GBRT model, error in the rescaling and error in the observational-driven products). In contrast to that, the error derived from the leave-one-model-out cross-validation approach illustrated in Figure 4a only shows the error in the statistical models themselves. We note that local relative errors can be very large (especially in regions with low absolute GPP) due to the uncertainty calculation method: Regardless of the absolute value, the error of each grid cell is at least 535 gC m<sup>-2</sup> yr<sup>-1</sup> (estimated SPE of the GBRT model using the RMSEP), which is added to the propagated errors of the observation-driven data sets and the error derived resulting from the rescaling (Equation 2). The GBRT model error is the dominant source of uncertainty. Since the true covariance structure of these global fields is unknown, a global (or at least regional) aggregation of these uncertainties is not possible.





**Figure 6.** Difference (a) and ratio (b) of the both biases shown in Figures 5b and 5c. For panel (b), all values close to 0 for the data set in the denominator have been masked to avoid divisions by 0. Both panels show that our approach is only to first-order bias correction, in which case both plots would only show constant values. (c) Comparison of the GBRT versus GBRT-1D projections of future GPP. This panel indicates a clear difference between the full GBRT using all predictors and the GBRT model using only historical GPP as single predictor. (d) Comparison of the GBRT versus LASSO projections of future GPP. This panel shows that there is a clear difference between using the nonlinear GBRT model and the linear LASSO model. The results of the LASSO-1D model are not shown here because they are very similar to LASSO.

To further compare our GBRT approach to other statistical models, we also fed the observation-driven data into the GBRT-1D, LASSO, and LASSO-1D models to get a constrained projection of the RCP 8.5 GPP. Figures 6c and 6d show the difference of the other models to the full GBRT approach. Both panels show values close to 0 for most of the globe (i.e., the difference between the GBRT and GBRT-1D/LASSO is small), indicating that the bias-correcting property of our approach is also present in the other regression models that use only the historical GPP as single predictors. Thus, we conclude that the bias correction of our GBRT model originates from the observation of the historical GPP, which is also supported by Figure 4b, which shows historical GPP as by far most important feature. On the contrary, second-order corrections originate from the observations in the remaining predictors. These second-order corrections are also visible as the nonzero values in Figures 6c and 6d. This indicates that using the nonlinear GBRT model with all features improves the final result. Moreover, the similar patterns in both panels demonstrate that the GBRT-1D model emulates the strong linear relation of the historical GPP to future GPP and performs equally well as the linear models. The global estimates of 167 Gt C yr<sup>-1</sup> (GBRT-1D), 163 Gt C yr<sup>-1</sup> (LASSO), and 162 Gt C yr<sup>-1</sup> (LASSO-1D) are all consistent with the global constraint of 171  $\pm$  12 Gt C yr<sup>-1</sup>.

#### 3.3. Constraining Gridded Fractional GPP Change Projections (Step 2b)

In Step 2b, we constrain the gridded fractional GPP change over the 21st century (2100 vs. 2000) in the emission-driven RCP 8.5 scenario (rescaled using Equation 2). This step is independent from Step 2a and only differs in the target variable (GPP change instead of absolute GPP). The remaining setup including the predictors, the GBRT model itself and the data sets are similar.





Figure 7. (a) Boxplot of the root-mean-square error of prediction (RMSEP) distributions for four different statistical models used to predict the fractional GPP change over the 21st century (Step 2b) using the leave-one-model-out crossvalidation approach. The distribution for each statistical model contains seven points (black dots, one for each climate model used as truth) and is represented in the following way: The lower and upper limit of the blue boxes correspond to the 25% and 75% quantiles, respectively. The central line in the box shows the median, the black "x" the mean of the distribution. The whiskers outside the box represent the range of the distribution. The GBRT algorithm shows the minimal mean and median RMSEP. Compared to the CMIP5 multimodel mean (MMM) and its corresponding rescaled version (rMMM), the prediction uncertainty measured by the mean RMSEP of the GBRT model is reduced by more than 16% and 9%, respectively. Compared to the linear LASSO model, the RMSEP mean and median of the GBRT model are reduced by 3%. (b) Relative global feature importance for the different features used in the LASSO model to predict the fractional GPP change (Step 2b). The feature importance for the LASSO model is given by the (normalized) linear coefficients of the model. (c) Local feature importance for the GBRT model used to predict fractional GPP change (Step 2b) calculated using the LIME approach (Ribeiro et al., 2016) for the three dominant features T (surface air temperature), LAI (leaf area index), and GPP. The relative influence of these three features (ignoring all other features) is color coded according to the triangle in the lower left corner: The vertices correspond to a single dominating feature, whereas the center corresponds to an equal influence of all three features. Over large parts of the globe, T is the dominant feature. (d) Relative global feature importance for the different features used in the GBRT model to predict the fractional GPP change (Step 2b). The blue bars for the two global feature importance plots (b) and (d) correspond to negative Pearson correlation coefficients between all predictors and the target variable.

# 3.3.1. Prediction Error in a Leave-One-Model-Out Cross-Validation Approach and Feature Importance

Figure 7a shows the RMSEP distributions for four different statistical models: the simple CMIP5 multimodel mean of the fractional GPP change (MMM), its rescaled version using Equation 2 (rMMM), a linear LASSO regression model and the GBRT model. Similar to Step 2a, the GBRT approach shows the smallest mean RMSEP values of all statistical models. Compared to MMM and rMMM the mean is reduced by 16% and 9%, respectively. Moreover, the nonlinear GBRT model also outperforms the linear LASSO model (mean reduced by 3%). In contrast to Step 2a, there is not a single predictor which heavily dominates the feature importance of the GBRT model. Thus, using single-predictor models (GBRT-1D and LASSO-1D) is not necessary in Step 2b. Figures 7b and 7d show the global feature importance for the LASSO and GBRT model, respectively. Both models agree that the surface air temperature (T) and the leaf area index (LAI) are the two





**Figure 8.** Fractional GPP change over the 21st century (2100 vs. 2000) in the RCP 8.5 scenario for different statistical models (Step 2b): (a) CMIP5 multimodel mean of the fractional GPP change (MMM), (b) rescaled CMIP5 multimodel mean using Equation 2 (rMMM), (c) linear LASSO model, and (d) GBRT model. The geographical patterns from the different statistical models are very similar showing a higher GPP increase in high latitudes and a lower GPP closer to the equator.

dominant predictors with a relative importance of more than 65%. The LASSO model shows a relative importance of less than 10% for each of the remaining features, while the GBRT model exhibits a considerably high relative importance for GPP with almost 20%. All predictors are negatively correlated to the target variable (as given by the sign of the linear Pearson correlation coefficient). That means cold areas with a small LAI in the historical climate show a larger GPP change over the 21st century and vice versa. Indeed, the fertilization effect is expected to be stronger in lower LAI regions. To get more detailed insights on this, observation-driven data is fed into the GBRT model to obtain a constrained fractional GPP change over the 21st century.

#### 3.3.2. Observation-Based GBRT Prediction of Relative GPP

The global distribution of the fractional GPP change for the different statistical models is shown in Figure 8. All panels show a GPP increase over the 21st century in the RCP 8.5 scenario for almost all regions of the globe. Compared to the unweighted CMIP5 multimodel mean (MMM), its rescaled version (rMMM) shows a slightly lower GPP increase, which is consistent with the emergent constraint approach from Step 1 that shows a lower global mean GPP increase over the 21st century (39%) than the global CMIP5 ensemble mean (43%). This is also illustrated in Figure 9a, which shows an almost constant pattern over the whole globe with a mean of 0.91 that is consistent with the ratio 39 % /43 %  $\approx$  0.91. Similar to Step 2a, can be interpreted as a global correction of the ESMs' response of future GPP to changes in the atmospheric CO<sub>2</sub> concentration.

All four patterns of Figure 8 show very similar geographical pattern. Obvious exceptions are the Sahara desert and the Arabian Peninsula, which show a noisy behavior for the multimodel means (MMM and rMMM) and a small GPP increase for the two regression models (LASSO and GBRT). This noise-like pattern occurs due to numerical inconsistencies produced by very small values in the absolute historical and future GPP in the climate models in this region that are used to derive the fractional GPP change following









**Figure 9.** (a) Ratio of the rescaled CMIP5 ensemble mean of the fractional GPP change over the 21st century using Equation 2 (rMMM, see Figure 8b) and its unweighted version (MMM, see Figure 8b). The plot shows an almost constant value over the whole globe with a mean of 0.91, which corresponds to the ratio of the constrained global mean GPP change over the 21st century (39%) and the CMIP5 ensemble mean global mean GPP change (43%) from Step 1. All values close to 0 for the data set in the denominator have been masked to avoid divisions by 0. Absolute (b) and relative (c) differences between the fractional GPP change over the 21st century predicted by the GBRT model (see Figure 8d) and rMMM (see Figure 8b). For panel (c), all values close to 0 for the data set in the denominator over large parts of the globe (corresponding to a value of 0). There are large absolute differences in the Sahara region and central Asia. The largest relative differences (except for the Sahara and Arabian Peninsula region) appear over South America, South Africa, the west coast of Africa, the Middle East, parts of Australia, and western parts of the United States.

Equation 1. However, due to the small impact on the global mean GPP this region is negligible in our analysis. The fractional GPP change is not uniformly distributed over the globe. Instead, there is a pronounced latitudinal dependency in its geographical pattern: in high latitudes, the projected GPP change is larger than in regions closer to the equator. This effect is particularly strong for the Northern high latitudes and consistent with the results of Wenzel, Cox, et al. (2016), who find an increased GPP change in northern high-latitude ecosystems compared to the northern extratropics. A possible reason for this is the extension of the growing season in high latitudes caused by climate change connected to a "greening," which has already been observed in the past climate through satellite measurements (Lucht et al., 2002; Myneni et al., 1997; Zhang et al., 2020) and associated with the increase in the CO<sub>2</sub> seasonal cycle amplitude (Forkel et al., 2016). This is consistent with the feature map in Figure 7c, which shows an increased relative importance of the LAI (a quantity directly related to the greening) in the high-latitude regions with high GPP changes. On the contrary, in the tropical ecosystems the growing season already covers the whole year. Thus, a further climate change-induced extension is not possible, leading to a smaller fractional GPP change in regions closer to the equator. For most parts of the remaining globe, the temperature (T) is the dominant predictor, with some areas in northern Africa, the Middle East, India and Australia showing an increased relative importance of the historical GPP. Overall, the negative correlation of the three most important features (T, LAI, and historical GPP) to the target variable is well reflected in the global pattern of the fractional GPP change: the high latitudes with lower T, lower LAI and lower



historical GPP show higher GPP changes, whereas regions closer to the equator with higher *T*, higher LAI and higher historical GPP show smaller GPP changes. At this point we want to emphasize that it is not possible to infer global or other large-scale averages of the fractional GPP change from the geographical distributions in Figure 8 since division operations and averaging operations do not commute: in general,  $\sum_i (x_i/y_i) \neq (\sum_i x_i)/(\sum_i y_i)$ . Thus, a direct comparison to other studies that give locally/globally aggregated results is not possible.

To get further insights in the GBRT algorithm, we compare it directly to the rescaled CMIP5 multimodel mean (rMMM), which is illustrated in Figures 9b and 9c. The mismatch between GBRT and rMMM can be interpreted as additional information that is added to the target variable by the observations in the predictors in Step 2b. Overall, the absolute difference (Figure 9b) is small over the whole globe, indicating that the rMMM and GBRT agree in the general pattern of the fractional GPP change. The most striking deviation in this panel is again the Sahara and Arabian Peninsula region, which shows the already discussed large absolute differences. Apart from that, there is a large patch in central Asia that shows high negative differences, that is, the GBRT model predicts a smaller GPP change in this area. The relative difference (Figure 9c) indicates many regions where the GBRT model predicts an increased GPP change compared to rMMM, in particular South America, South Africa, the west coast of Africa, the Middle East, parts of Australia and western parts of the United States. However, all these correspond to low absolute GPP changes. Figure S6 shows the corresponding gridded uncertainties for all four statistical models that also cover the uncertainty in the emergent constraint and the observational uncertainty (in contrast to Figure 7a, which only covers the uncertainty in the statistical models). Similar to Step 2a, the local standard errors are large, even for the GBRT model (that performs better than any other statistical model) which shows errors of at least 43.6% for every grid cell caused by the uncertainty in the statistical model itself. Figure S2b illustrates the residuals for the training and independent test data sets and shows that the GBRT model has symmetrical residuals and is not overfitting.

#### 3.4. Comparison of the Absolute and Relative GBRT Prediction

In a last step, we use the FLUXNET-MTE observational product to constrain the absolute GPP increase at the end of the 21st century (2091–2100) from the fractional GPP change given by the GBRT model in Step 2b. This can be directly compared to the output of Step 2a and serves as a sanity check between Steps 2a and 2b. As shown by Figure 10, both approaches show similar geographical patterns (pattern correlation  $R^2 = 0.97$ ). The difference between the two approaches is in the same order of magnitude as the difference between the difference between the difference (see Figures 6c and 6d) and considerably smaller than the difference between the GBRT and rMMM approaches (see Figure 5b). Moreover, the globally aggregated results (169 Gt C yr<sup>-1</sup> for Step 2a and 175 Gt C yr<sup>-1</sup> for Step 2b) are both compatible with the global value of  $171 \pm 12$  Gt C yr<sup>-1</sup> given by the global emergent constraint in Step 1.

## 4. Summary and Discussion

In this paper, we developed a two-step approach to constrain the projected GPP at the end of the 21st century (2091-2100) in the RCP 8.5 scenario. In the first step, we constrain the global mean GPP to  $171 \pm 12$  Gt C yr<sup>-1</sup> using a published emergent constraint approach which we assume can also be applied to the global mean GPP at the end of the 21st century in the RCP 8.5 scenario (see discussion in section 2.1). This step corresponds to the correction of the ESM's biases in the response of future GPP to rising atmospheric CO<sub>2</sub> concentration, which is the main source of uncertainty in future GPP projections (Arora et al., 2013; Haverd et al., 2020; Rogers et al., 2017). As already discussed in section 2.1, due to other than CO<sub>2</sub> fertilization drivers of the increase in the CO<sub>2</sub> seasonal cycle amplitude (Bastos et al., 2019; Forkel et al., 2016; Piao et al., 2018; Zhao et al., 2016) this emergent constraint is not undisputed and can be replaced with another emergent constraint in future studies. In the second step, a machine learning approach is used to further constrain the gridded GPP based on process-based present-day predictors. We consider two target variables: first (Step 2a) the gridded monthly climatologies of absolute GPP (2091-2100) and second (Step 2b) the gridded fractional GPP change over the 21st century (years 2091-2100 vs. years 1991-2000, see Equation 1). Both approaches give consistent results (see Figure 10). The latter quantity shows an increased GPP change in the high latitudes compared to regions closer to the equator, which might be attributed to an additional greening trend in the high latitudes as supported by the feature map in Figure 7c that shows the





**Figure 10.** (top row) Absolute future GPP at the end of the 21st century (2091–2100) calculated using (a) Step 2a and (b) Step 2b. Both approaches give similar results, with global averages of (a) 169 Gt C yr<sup>-1</sup> and (b) 175 Gt C yr<sup>-1</sup>, which are both consistent with the global result of Step 1 (171 ± 12 Gt C yr<sup>-1</sup>). The pattern correlation between both approaches is  $R^2 = 0.97$ . (bottom row) Absolute (c) and relative (d) differences between panels (a) and (b).

LAI as dominant predictor in this region. However, in the absence of a robust physical mechanism this connection cannot be proven. For both target variables, the GBRT algorithm is superior to all other considered statistical models in terms of prediction uncertainty evaluated in a leave-one-model-out crossvalidation approach among different statistical models. Compared to the unweighted CMIP5 multimodel mean the mean prediction error is reduced by 48% (Step 2a) and 16% (Step 2b); compared to a linear LASSO regression model the mean prediction error is reduced by approximately 3% for both cases. However, local standard errors are still large for both target variables, even for the GBRT model. Due to the unknown covariance structure a global aggregation of these errors is not possible. Step 2a mainly corrects the bias of the simulated absolute historical GPP relative to observations (Anav et al., 2013). Consequently, the historical GPP is by far the most important predictor for future absolute GPP (regions with high GPP in the past are likely to have high GPP in the future and vice versa). However, as we have shown in Figure 6, the GBRT model expands this bias correction by taking more predictors than the historical GPP into account. A similar result is provided by Step 2b: Figures 9b and 9c show the impact of the additional predictors as difference between the CMIP5 multimodel mean and the GBRT predicted GPP change. In this case, the target variable (fractional GPP change) was already normalized with the historical GPP. Therefore, the temperature (second most important feature in Step 2a) is now the most important predictor instead of historical GPP (see Figure 7d). As shown in Figure 7c, the different features have also different dominant regional impacts; but again, since there is no distinct physical mechanism relating the predictors to the target variable this map needs further analysis and treated with care.

Our GBRT approach (Steps 2) is mathematically similar to MDER (Karpechko et al., 2013; Senftleben et al., 2019; Wenzel, Eyring, et al., 2016): we establish a relationship between process-oriented, physically relevant diagnostics and future projections and then utilize this to predict today's observed conditions into



the future. We emphasize that the exact nature of this relationship is strongly dependent on the climate model ensemble considered. Although derived empirically, we argue that this relationship can still be used to extract further information from the climate model ensemble for two reasons: First, we consider process-oriented variables which are physically linked to GPP, and, second, we ensure its statistical robustness. The latter is achieved by considering gridded climatological data instead of global/regional means. The number of points establishing the relation is dramatically increased from about 50 points for classical emergent constraints to 237'852 points (Step 2a) and 16'503 (Step 2b) in our approach. We validate the relationship by extensive out-of-sample testing using a leave-one-model-out cross-validation approach and randomly excluding parts of the data prior to training. Furthermore, the large number of points in combination with the nonlinear GBRT model allows us to exploit more information than classical emergent constraints and include nonlinear multivariate relations in the statistical model, including the information encoded in spatial variation. The prediction phase of our approach (feeding observation-based data into the trained GBRT model) can be interpreted as an implicit performance weighting: the GBRT model creates predictions based on today's conditions (in form of variables which are physically connected to GPP). However, unlike other performance weighting techniques (Knutti et al., 2017; Sanderson et al., 2017), we do not assign constant weights to the different climate models. On the contrary, since the algorithm works on grid cell level, it is able to adapt to regional climate characteristics and thus applies an implicit "local" and "climate dependent" weighting. Moreover, the algorithm adds another level of weighting by assigning a relative relevance score to the different predictors. We emphasize that this is only an implicit weighting, since it is not possible to extract specific values for the individual weights due to the complex structure of the GBRT model.

Yet, like emergent constraints and performance weighting techniques, our GBRT approach in Step 2 hinges on the assumption that the climate models reflect the real world. This is certainly only partially true, since there are processes like  $CO_2$  fertilization, land use and land cover changes, nutrient cycles and limitations (Du et al., 2020; Fleischer et al., 2019), disturbances, and induced vegetation dynamics which may strongly alter the future trajectories of GPP without this being fully encoded in today's observation products. Moreover, all  $CO_2$ -related effects are only implicitly incorporated in our approach by the target period we are using and the global rescaling using the emergent constraint in Step 1. Thus, our reduction of uncertainty only applies to the specific setup we are considering here: the future GPP in the CMIP5 RCP 8.5 scenario at the end of the 21st century. It has to be viewed as a nominal reduction, while the real uncertainty remains unknown and is likely larger than the nominal one. These issues could be tested in the future by including other (even offline) models, which account for the aforementioned processes. A true validation will only be possible once we experience the changed condition or could be based on paleo data. In summary, our approach does not address the coupled system directly but hinges on the climate predictions of the individual ESMs and hence is not able to implicitly correct respective biases of the CMIP5 ensemble.

The presented approach based on machine learning is not limited to projections of the future carbon cycle, which was used as an illustrative example. Indeed, since its only prerequisites are the availability of gridded climate model data and gridded observation-based data, the GBRT algorithm can be applied to any variable of interest if physically relevant diagnostics that influence the target variable are known. This opens a wide range of possibilities for constraining uncertainties in projected variable scenarios. Additionally, maps of the local relative feature importance as shown in Figure 7c can be used to reveal connections between different Earth system variables that are currently unknown. With the proposed data-driven approach, a possibly extended application could be to constrain projections of not just one but several variables simultaneously. Moreover, the concept of our method is not limited to GBRT but can be used with any other (machine learning) regression algorithm.

## Data Availability Statement

The corresponding ESMValTool recipe that can be used to reproduce the figures of this paper is included in ESMValTool v2.1 (Eyring et al., 2020; Lauer et al., 2020; Righi et al., 2020). ESMValTool v2.1 is released under the Apache License, VERSION 2.0. The latest release of the ESMValTool is publicly available on Zenodo at https://doi.org/10.5281/zenodo.3401363. The source code of the ESMValCore package, which is installed as a dependency of the ESMValTool, is also publicly available on Zenodo at https://doi.org/



10.5281/zenodo.3387139. ESMValTool and ESMValCore are developed on the GitHub repositories available at https://github.com/ESMValGroup. All scripts are implemented in Python and use the following packages for their core functionalities: sklearn (Pedregosa et al., 2011), xgboost (Chen & Guestrin, 2016), and lime (Ribeiro et al., 2016). CMIP5 model output (see Table S1) is available through the official nodes of the Earth System Grid Foundation (ESGF) and can be directly used within the ESMValTool (e.g., https://esgf-data.dkrz.de/projects/esgf-dkrz/). Download instructions and preprocessing scripts for observation-driven products (CRU, ERA-Interim, LAI3g, FLUXNET-MTE, and Scripps-CO2-KUM) are included in the ESMValTool.

#### Acknowledgments

The collaboration on this paper was initiated by V. E., M. R., G. C. V., and P. G. as part of the proposal writing of the European Research Council (ERC) Synergy Grant "Understanding and Modelling the Earth System with Machine Learning (USMILE)" under Grant Agreement No 855187. M. S. and P. F. were supported by the European Union's Horizon 2020 project "Coordinated Research in Earth Systems and Climate: Experiments, kNowledge, Dissemination and Outreach (CRESCENDO)" under Grant Agreement No. 641816 and the European Union's Horizon 2020 project "Climate-Carbon Interactions in the Coming Century" (4C) under Grant Agreement No. 821003. G. C. V. was supported by the ERC Consolidator Grant "Statistical Learning for Earth Observation Data Analysis (SEDAL)" under Grant Agreement No. 647423. We acknowledge the World Climate Research Program's (WCRP's) Working Group on Coupled Modeling (WGCM), which is responsible for CMIP, and we thank the climate modeling groups for producing and making available their model output, the Earth System Grid Federation (ESGF) for archiving the data and providing access, and the multiple funding agencies who support CMIP and ESGF. The computational resources of the Deutsches Klima Rechenzentrum (DKRZ, Hamburg, Germany) where the ESMValTool is fully integrated into the ESGF infrastructure are kindly acknowledged.

#### References

- Allen, M. R., & Ingram, W. J. (2002). Constraints on future changes in climate and the hydrologic cycle. *Nature*, 419(6903), 224. https://doi. org/10.1038/nature01092
- Anav, A., Friedlingstein, P., Kidston, M., Bopp, L., Ciais, P., Cox, P., et al. (2013). Evaluating the land and ocean components of the global carbon cycle in the CMIP5 Earth system models. *Journal of Climate*, 26(18), 6801–6843. https://doi.org/10.1175/jcli-d-12-00417.1
- Arora, V. K., Boer, G. J., Friedlingstein, P., Eby, M., Jones, C. D., Christian, J. R., et al. (2013). Carbon-concentration and carbon-climate feedbacks in CMIP5 Earth system models. *Journal of Climate*, 26(15), 5289–5314. https://doi.org/10.1175/jcli-d-12-00494.1
- Bastos, A., Ciais, P., Chevallier, F., Rodenbeck, C., Ballantyne, A. P., Maignan, F., et al. (2019). Contrasting effects of CO<sub>2</sub> fertilization, land-use change and warming on seasonal amplitude of Northern Hemisphere CO<sub>2</sub> exchange. *Atmospheric Chemistry and Physics*, 19(19), 12,361–12,375. https://doi.org/10.5194/acp-19-12361-2019
- Bishop, C. M. (2006). Pattern recognition and machine learning. New York, NY: Springer.
- Bodman, R., Rayner, P. J., & Karoly, D. J. (2013). Uncertainty in temperature projections reduced using carbon cycle and climate observations. *Nature Climate Change*, 3(8), 725–729. https://doi.org/10.1038/nclimate1903
- Booth, B. B., Jones, C. D., Collins, M., Totterdell, I. J., Cox, P. M., Sitch, S., et al. (2012). High sensitivity of future global warming to land carbon cycle processes. *Environmental Research Letters*, 7(2). https://doi.org/10.1088/1748-9326/7/2/024002
- Chen, T., & Guestrin, C. (2016). XGBoost: A scalable tree boosting system (pp. 785–794). Paper presented at Proceedings of the 22nd ACM SIGKDD International Conference on Knowledge Discovery and Data Mining, Association for Computing Machinery, San Francisco, CA. https://doi.org/10.1145/2939672.2939785
- Ciais, P., Sabine, C., Bala, G., Bopp, L., Brovkin, V., Canadell, J., et al. (2013). Carbon and other biogeochemical cycles. In Climate change 2013: The physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 465–570). Cambridge, UK and New York, NY: Cambridge University Press.
- Collins, M., Knutti, R., Arblaster, J., Dufresne, J.-L., Fichefet, T., & Friedlingstein, P. (2013). Long-term climate change: Projections, commitments and irreversibility. In *Climate change 2013: The physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change* (pp. 1029–1136). Cambridge, UK and New York, NY: Cambridge University Press.
- Cox, P. M., Pearson, D., Booth, B. B., Friedlingstein, P., Huntingford, C., Jones, C. D., & Luke, C. M. (2013). Sensitivity of tropical carbon to climate change constrained by carbon dioxide variability. *Nature*, 494(7437), 341–344. https://doi.org/10.1038/nature11882
- Crafts-Brandner, S. J., & Salvucci, M. E. (2000). Rubisco activase constrains the photosynthetic potential of leaves at high temperature and CO<sub>2</sub>. *Proceedings of the National Academy of Sciences of the United States of America*, 97(24), 13,430–13,435. https://doi.org/10.1073/pnas.230451497
- de Elia, R., Laprise, R., & Denis, B. (2002). Forecasting skill limits of nested, limited-area models: A perfect-model approach. *Monthly Weather Review*, 130(8), 2006–2023. https://doi.org/10.1175/1520-0493(2002)130<2006:fslonl>2.0.co;2
- De'ath, G. (2007). Boosted trees for ecological modeling and prediction. *Ecology*, 88(1), 243–251. https://doi.org/10.1890/0012-9658(2007)88 [243:btfema]2.0.co;2
- Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-Interim reanalysis: Configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society*, 137(656), 553–597. https://doi.org/ 10.1002/qj.828
- Du, E. Z., Terrer, C., Pellegrini, A. F. A., Ahlstrom, A., van Lissa, C. J., Zhao, X., et al. (2020). Global patterns of terrestrial nitrogen and phosphorus limitation. *Nature Geoscience*, 13(3), 221. https://doi.org/10.1038/s41561-019-0530-4
- Elith, J., Leathwick, J. R., & Hastie, T. (2008). A working guide to boosted regression trees. *Journal of Animal Ecology*, 77(4), 802–813. https://doi.org/10.1111/j.1365-2656.2008.01390.x
- Eyring, V., Bock, L., Lauer, A., Righi, M., Schlund, M., Andela, B., et al. (2020). Earth System Model Evaluation Tool (ESMValTool) v2.0 an extended set of large-scale diagnostics for quasi-operational and comprehensive evaluation of Earth system models in CMIP. *Geoscientific Model Development*, *13*(7), 3383–3438. https://doi.org/10.5194/gmd-13-3383-2020
- Eyring, V., Cox, P. M., Flato, G. M., Gleckler, P. J., Abramowitz, G., Caldwell, P., et al. (2019). Taking climate model evaluation to the next level. *Nature Climate Change*, 9(2), 102–110. https://doi.org/10.1038/s41558-018-0355-y
- Fleischer, K., Rammig, A., De Kauwe, M. G., Walker, A. P., Domingues, T. F., Fuchslueger, L., et al. (2019). Amazon forest response to CO<sub>2</sub> fertilization dependent on plant phosphorus acquisition. *Nature Geoscience*, *12*(9), 736. https://doi.org/10.1038/s41561-019-0404-9
- Forkel, M., Carvalhais, N., Rodenbeck, C., Keeling, R., Heimann, M., Thonicke, K., et al. (2016). Enhanced seasonal CO<sub>2</sub> exchange caused by amplified plant productivity in northern ecosystems. *Science*, 351(6274), 696–699. https://doi.org/10.1126/science.aac4971
- Freund, Y., & Schapire, R. E. (1996). *Experiments with a new boosting algorithm* (pp. 148–156). Paper presented at Proceedings of the Thirteenth International Conference on International Conference on Machine Learning, Morgan Kaufmann Publishers Inc., San Francisco, CA. Retrieved from https://www.cis.upenn.edu/~mkearns/teaching/COLT/boostingexperiments.pdf
- Friedlingstein, P., Cox, P., Betts, R., Bopp, L., Von Bloh, W., Brovkin, V., et al. (2006). Climate-carbon cycle feedback analysis: Results from the (CMIP)-M-4 model intercomparison. *Journal of Climate*, 19(14), 3337–3353. https://doi.org/10.1175/jcli3800.1
- Friedlingstein, P., Jones, M. W., O'Sullivan, M., Andrew, R. M., Hauck, J., Peters, G. P., et al. (2019). Global Carbon Budget 2019. Earth System Science Data, 11(4), 1783–1838. https://doi.org/10.5194/essd-11-1783-2019

Friedman, J. H. (2001). Greedy function approximation: A gradient boosting machine. The Annals of Statistics, 29(5), 1189–1232. https:// doi.org/10.1214/aos/1013203451

Friedman, J. H. (2002). Stochastic gradient boosting. Computational Statistics & Data Analysis, 38(4), 367–378. https://doi.org/10.1016/ s0167-9473(01)00065-2

Graven, H. D., Keeling, R. F., Piper, S. C., Patra, P. K., Stephens, B. B., Wofsy, S. C., et al. (2013). Enhanced seasonal exchange of CO<sub>2</sub> by northern ecosystems since 1960. *Science*, 341(6150), 1085–1089. https://doi.org/10.1126/science.1239207

Gray, J. M., Frolking, S., Kort, E. A., Ray, D. K., Kucharik, C. J., Ramankutty, N., & Friedl, M. A. (2014). Direct human influence on atmospheric CO<sub>2</sub> seasonality from increased cropland productivity. *Nature*, *515*(7527), 398. https://doi.org/10.1038/nature13957

Gregory, J. M., Jones, C. D., Cadule, P., & Friedlingstein, P. (2009). Quantifying carbon cycle feedbacks. *Journal of Climate*, 22(19), 5232–5250. https://doi.org/10.1175/2009JCLI2949.1

Harris, I., Jones, P. D., Osborn, T. J., & Lister, D. H. (2014). Updated high-resolution grids of monthly climatic observations—The CRU TS3.10 dataset. International Journal of Climatology, 34(3), 623–642. https://doi.org/10.1002/joc.3711

Haverd, V., Smith, B., Canadell, J. G., Cuntz, M., Mikaloff-Fletcher, S., Farquhar, G., et al. (2020). Higher than expected CO<sub>2</sub> fertilization inferred from leaf to global observations. *Global Change Biology*, 26. https://doi.org/10.1111/gcb.14950

Huntzinger, D. N., Michalak, A. M., Schwalm, C., Ciais, P., King, A. W., Fang, Y., et al. (2017). Uncertainty in the response of terrestrial carbon sink to environmental drivers undermines carbon-climate feedback predictions. *Scientific Reports*, 7. https://doi.org/10.1038/ s41598-017-03818-2

Jung, M., Reichstein, M., Margolis, H. A., Cescatti, A., Richardson, A. D., Arain, M. A., et al. (2011). Global patterns of land-atmosphere fluxes of carbon dioxide, latent heat, and sensible heat derived from eddy covariance, satellite, and meteorological observations. *Journal* of Geophysical Research, 116, G00J07. https://doi.org/10.1029/2010JG001566

Karpechko, A. Y., Maraun, D., & Eyring, V. (2013). Improving Antarctic total ozone projections by a process-oriented multiple diagnostic ensemble regression. Journal of the Atmospheric Sciences, 70(12), 3959–3976. https://doi.org/10.1175/jas-d-13-071.1

Keeling, C. D., Chin, J. F. S., & Whorf, T. P. (1996). Increased activity of northern vegetation inferred from atmospheric CO<sub>2</sub> measurements. *Nature*, 382(6587), 146–149. https://doi.org/10.1038/382146a0

Keeling, C. D., Piper, S. C., Bacastow, R. B., Wahlen, M., Whorf, T. P., Heimann, M., & Meijer, H. A. (2005). Atmospheric CO<sub>2</sub> and <sup>13</sup>CO<sub>2</sub> exchange with the terrestrial biosphere and oceans from 1978 to 2000: Observations and carbon cycle implications. In A history of atmospheric CO<sub>2</sub> and its effects on plants, animals, and ecosystems (pp. 83–113). New York, NY: Springer.

Keeling, C. D., Whorf, T. P., Wahlen, M., & Vanderplicht, J. (1995). Interannual extremes in the rate of rise of atmospheric carbon-dioxide since 1980. Nature, 375(6533), 666–670. https://doi.org/10.1038/375666a0

Knutti, R., Sedlacek, J., Sanderson, B. M., Lorenz, R., Fischer, E. M., & Eyring, V. (2017). A climate model projection weighting scheme accounting for performance and interdependence. *Geophysical Research Letters*, 44, 1909–1918. https://doi.org/10.1002/ 2016GL072012

Lauer, A., Eyring, V., Bellprat, O., Bock, L., Gier, B. K., Hunter, A., et al. (2020). Earth System Model Evaluation Tool (ESMValTool) v2.0 – diagnostics for emergent constraints and future projections from Earth system models in CMIP. *Geoscientific Model Development*, 13(9), 4205–4228. https://doi.org/10.5194/gmd-13-4205-2020

Lian, X., Piao, S. L., Huntingford, C., Li, Y., Zeng, Z. Z., Wang, X. H., et al. (2018). Partitioning global land evapotranspiration using CMIP5 models constrained by observations. *Nature Climate Change*, 8(7), 640. https://doi.org/10.1038/s41558-018-0207-9

Lucht, W., Prentice, I. C., Myneni, R. B., Sitch, S., Friedlingstein, P., & Cramer, W. (2002). Climatic control of the high-latitude vegetation greening trend and Pinatubo effect. Science, 296(5573), 1687–1689. https://doi.org/10.1126/science.1071828

Matthews, H. D., Gillett, N. P., Stott, P. A., & Zickfeld, K. (2009). The proportionality of global warming to cumulative carbon emissions. *Nature*, 459(7248), 829–U823. https://doi.org/10.1038/nature08047

Myneni, R. B., Keeling, C. D., Tucker, C. J., Asrar, G., & Nemani, R. R. (1997). Increased plant growth in the northern high latitudes from 1981 to 1991. Nature, 386(6626), 698–702. https://doi.org/10.1038/386698a0

Pedregosa, F., Varoquaux, G., Gramfort, A., Michel, V., Thirion, B., Grisel, O., et al. (2011). Scikit-learn: Machine learning in Python.

Journal of Machine Learning Research, 12, 2825–2830. https://jmlr.csail.mit.edu/papers/v12/pedregosa11a.html

Piao, S. L., Liu, Z., Wang, Y. L., Ciais, P., Yao, Y. T., Peng, S., et al. (2018). On the causes of trends in the seasonal amplitude of atmospheric CO<sub>2</sub>. Global Change Biology, 24(2), 608–616. https://doi.org/10.1111/gcb.13909

Riahi, K., Rao, S., Krey, V., Cho, C. H., Chirkov, V., Fischer, G., et al. (2011). RCP 8.5—A scenario of comparatively high greenhouse gas emissions. *Climatic Change*, 109(1–2), 33–57. https://doi.org/10.1007/s10584-011-0149-y

Ribeiro, M. T., Singh, S., & Guestrin, C. (2016). "Why Should I Trust You?": Explaining the predictions of any classifier (pp. 1135–1144). San Francisco, CA: Proceedings of the 22nd ACM SIGKDD International Conference on Knowledge Discovery and Data Mining, Association for Computing Machinery. https://doi.org/10.1145/2939672.2939778

Righi, M., Andela, B., Eyring, V., Lauer, A., Predoi, V., Schlund, M., et al. (2020). Earth system model evaluation tool (ESMValTool) v2.0— Technical overview. *Geoscientific Model Development*, 13(3), 1179–1199. https://doi.org/10.5194/gmd-13-1179-2020

Rogers, A., Medlyn, B. E., Dukes, J. S., Bonan, G., von Caemmerer, S., Dietze, M. C., et al. (2017). A roadmap for improving the representation of photosynthesis in Earth system models. *New Phytologist*, 213(1), 22–42. https://doi.org/10.1111/nph.14283

Sanderson, B. M., Wehner, M., & Knutti, R. (2017). Skill and independence weighting for multi-model assessments. Geoscientific Model Development, 10(6), 2379–2395. https://doi.org/10.5194/gmd-10-2379-2017

Senftleben, D., Lauer, A., & Karpechko, A. (2019). Constraining uncertainties in CMIP5 projections of September Arctic sea ice extent with observations. Journal of Climate, 33. https://doi.org/10.1175/JCLI-D-19-0075.1

Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498. https://doi.org/10.1175/bams-d-11-00094.1

Thomas, R. T., Prentice, L. C., Graven, H., Ciais, P., Fisher, J. B., Hayes, D. J., et al. (2016). Increased light-use efficiency in northern terrestrial ecosystems indicated by CO<sub>2</sub> and greening observations. *Geophysical Research Letters*, 43, 11,339–11,349. https://doi.org/ 10.1002/2016GL070710

Tibshirani, R. (1996). Regression shrinkage and selection via the Lasso. Journal of the Royal Statistical Society: Series B: Methodological, 58(1), 267–288. https://doi.org/10.1111/j.2517-6161.1996.tb02080.x

UNFCCC. (2015). Adoption of the Paris Agreement (FCCC/CP/2015/L.9/Rev.1). Paris: United Nations Framework Convention on Climate Change (UNFCCC). Retrieved from http://unfccc.int/resource/docs/2015/cop21/eng/l09r01.pdf

Walker, A. P., De Kauwe, M. G., Bastos, A., Belmecheri, S., Georgiou, K., Keeling, R., et al. (2020). Integrating the evidence for a terrestrial carbon sink caused by increasing atmospheric CO<sub>2</sub>. New Phytologist, 1. https://doi.org/10.1111/nph.16866



- Wenzel, S., Cox, P. M., Eyring, V., & Friedlingstein, P. (2014). Emergent constraints on climate-carbon cycle feedbacks in the CMIP5 Earth system models. Journal of Geophysical Research: Biogeosciences, 119, 794–807. https://doi.org/10.1002/2013JG002591
- Wenzel, S., Cox, P. M., Eyring, V., & Friedlingstein, P. (2016). Projected land photosynthesis constrained by changes in the seasonal cycle of atmospheric CO<sub>2</sub>. Nature, 538(7626), 499. https://doi.org/10.1038/nature19772
- Wenzel, S., Eyring, V., Gerber, E. P., & Karpechko, A. Y. (2016). Constraining future summer austral jet stream positions in the CMIP5 ensemble by process-oriented multiple diagnostic regression. *Journal of Climate*, 29(2), 673–687. https://doi.org/10.1175/JCLI-D-15-0412.1
- Winkler, A. J., Myneni, R. B., Alexandrov, G. A., & Brovkin, V. (2019). Earth system models underestimate carbon fixation by plants in the high latitudes. *Nature Communications*, 10. https://doi.org/10.1038/s41467-019-08633-z
- Winkler, A. J., Myneni, R. B., & Brovkin, V. (2019). Investigating the applicability of emergent constraints. Earth System Dynamics, 10(3), 501–523. https://doi.org/10.5194/esd-10-501-2019
- Zhang, Y., Parazoo, N. C., Williams, A. P., Zhou, S., & Gentine, P. (2020). Large and projected strengthening moisture limitation on end-ofseason photosynthesis. Proceedings of the National Academy of Sciences of the United States of America, 117(17), 9216–9222. https://doi. org/10.1073/pnas.1914436117
- Zhao, F., & Zeng, N. (2014). Continued increase in atmospheric CO<sub>2</sub> seasonal amplitude in the 21st century projected by the CMIP5 Earth system models. *Earth System Dynamics*, 5(2), 423–439. https://doi.org/10.5194/esd-5-423-2014
- Zhao, F., Zeng, N., Asrar, G., Friedlingstein, P., Ito, A., Jain, A., et al. (2016). Role of CO<sub>2</sub>, climate and land use in regulating the seasonal amplitude increase of carbon fluxes in terrestrial ecosystems: A multimodel analysis. *Biogeosciences*, 13, 5121–5137. https://doi.org/ 10.5194/bg-13-5121-2016
- Zhu, Z. C., Bi, J., Pan, Y. Z., Ganguly, S., Anav, A., Xu, L., et al. (2013). Global data sets of vegetation leaf area index (LAI)3g and fraction of photosynthetically active radiation (FPAR)3g derived from Global Inventory Modeling and Mapping Studies (GIMMS) normalized difference vegetation index (NDVI3g) for the period 1981 to 2011. *Remote Sensing*, 5(2), 927–948. https://doi.org/10.3390/rs5020927

## **References From the Supporting Information**

- Arora, V. K., Scinocca, J. F., Boer, G. J., Christian, J. R., Denman, K. L., Flato, G. M., et al. (2011). Carbon emission limits required to satisfy future representative concentration pathways of greenhouse gases. *Geophysical Research Letters*, 38, L05805. https://doi.org/10.1029/ 2010GL046270
- Collins, W. J., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Halloran, P., Hinton, T., et al. (2011). Development and evaluation of an Earth-system model-HadGEM2. *Geoscientific Model Development*, 4(4), 1051–1075. https://doi.org/10.5194/gmd-4-1051-2011
- Dunne, J. P., John, J. G., Adcroft, A. J., Griffies, S. M., Hallberg, R. W., Shevliakova, E., et al. (2012). GFDL's ESM 2 global coupled climate-carbon Earth system models. Part I: Physical formulation and baseline simulation characteristics. *Journal of Climate*, 25(19), 6646–6665. https://doi.org/10.1175/jcli-d-12-00150.1
- Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., et al. (2011). The Community Climate System Model Version 4. Journal of Climate, 24(19), 4973–4991. https://doi.org/10.1175/2011JCLI4083.1
- Giorgetta, M. A., Jungclaus, J., Reick, C. H., Legutke, S., Bader, J., Bottinger, M., et al. (2013). Climate and carbon cycle changes from 1850 to 2100 in MPI-ESM simulations for the Coupled Model Intercomparison Project Phase 5. Journal of Advances in Modeling Earth Systems, 5(3), 572–597. https://doi.org/10.1002/jame.20038
- Iversen, T., Bentsen, M., Bethke, I., Debernard, J. B., Kirkevag, A., Seland, O., et al. (2013). The Norwegian Earth System Model, NorESM1-M—Part 2: Climate response and scenario projections. *Geoscientific Model Development*, 6(2), 389–415. https://doi.org/10.5194/gmd-6-389-2013
- Watanabe, S., Hajima, T., Sudo, K., Nagashima, T., Takemura, T., Okajima, H., et al. (2011). MIROC-ESM 2010: Model description and basic results of CMIP5-20c3m experiments. *Geoscientific Model Development*, 4(4), 845–872. https://doi.org/10.5194/gmd-4-845-2011